\cleardoublepage
\begin{appendix}

\pagestyle{fancy}
\chapter{Testfiles}\label{appendix.testfiles}

In this thesis, several examples where used in the context of chord recognition\index{chord recognition} and onset detection\index{onset detection}. Table \ref{appendix.table_chord_files} gives an overview of the files used for chord recognition\index{chord recognition} and Table \ref{appendix.table_onset_files} lists the files used for onset detection\index{onset detection}. The spectrograms of the testfiles for chord recognition are given in Figures \ref{appendix.figure_chord_files1} and \ref{appendix.figure_chord_files2} and the spectrograms of the testfiles for onset detection in Figures \ref{appendix.figure_onset_files1} to \ref{appendix.figure_onset_files4}.

\begin{table}[ht]
	\centering
	\caption[Testfiles for chord recognition.]{Testfiles for chord recognition\index{chord recognition}.}
	\ \\[0.3cm]
	\begin{tabular}{clr}
		& & Duration \\
		Number & Title & [s] \\
		\hline
		1 & All My Loving 			& 25 \\
		2 & All You Need Is Love 	& 30 \\
		3 & And I Love Her 			& 29 \\
		4 & Baby's In Black 			& 37 \\
		5 & Back In The USSR 		& 30 \\
		6 & Cry Baby Cry 			& 30 \\
		7 & Helter Skelter 			& 29 \\
		8 & I Am The Walrus 		& 30 \\
		9 & I'm Only Sleeping 		& 34 \\
		10 & Let It Be 				& 26 \\
		11 & Savoy Truffle 			& 29 \\
		12 & Tell Me Why 			& 29 \\
		13 & You Can't Do That	 	& 29
	\end{tabular}
	\label{appendix.table_chord_files}
\end{table}

\begin{figure}[htb]
	\centering
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n1.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n2.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n3.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n4.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n5.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n6.eps}
	\caption[Spectrograms of testfiles for chord recognition (Testfiles 1 to 6).]{Spectrograms of testfiles for chord recognition. \textbf{Left to right, top to bottom:} Testfiles 1 to 6, compare Table \ref{appendix.table_chord_files}.}
	\label{appendix.figure_chord_files1}
\end{figure}
\begin{figure}[htb]
	\centering
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n7.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n8.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n9.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n10.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n11.eps}
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n12.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/chord/original/n13.eps}
	\caption[Spectrograms of testfiles for chord recognition (Testfiles 7 to 13).]{Spectrograms of testfiles for chord recognition. \textbf{Left to right, top to bottom:} Testfiles 7 to 13, compare Table \ref{appendix.table_chord_files}.}
	\label{appendix.figure_chord_files2}
\end{figure}

\begin{landscape}
\vfill
\begin{table}
	\centering
	\caption[Testfiles for onset detection.]{Testfiles for onset detection\index{onset detection}.}
%	\ \\[0.3cm]
	\begin{tabular}{clrl}
		& & Duration \\
		Number & Title & [s] & Remarks \\
		\hline
		1 & Friedrich Burgm{\"u}ller: Piano etude Op.\ 100, No.\ 2 	& 17 & l'Arabesque \\
		2 & Ludwig van Beethoven: 5th Symphony 						& 21 & Bernstein interpretation \\
		3 & Johannes Brahms: Hungarian Dance Op.\ 39, No.\ 5 		& 15 \\
		4 & Happy Birthday 											& 21 & Choir accompanied by a piano \\
		5 & In Frankrijk buiten de poorten 							& 8 & Dutch folk song, female singing, see \cite{dutchsongdatabase} \\
		6 & Ludwig van Beethoven: 5th Symphony 						& 21 & MIDI file \\
		7 & Alexander Borodin: String Quartet No.\ 2 				& 38 & Notturno \\
		8 & Dmitri Shostakovich: Waltz No.\ 2 						& 30 & Jazz Suite No.\ 2, Yablonsky interpretation \\
		9 & cello1													& 14 \\%& See also \cite{leveau04} \\
		10 & clarinet1												& 30 \\%& See also \cite{leveau04} \\
		11 & classic2												& 20 \\%& See also \cite{leveau04} \\
		12 & classic3												& 14 \\%& See also \cite{leveau04} \\
		13 & distguit1												& 6 \\%& See also \cite{leveau04} \\
		14 & guitar2												& 15 \\%& See also \cite{leveau04} \\
		15 & guitar3												& 15 \\%& See also \cite{leveau04} \\
		16 & jazz2													& 14 \\%& See also \cite{leveau04} \\
		17 & jazz3													& 11 \\%& See also \cite{leveau04} \\
		18 & piano1													& 15 \\%& See also \cite{leveau04} \\
		19 & pop1													& 15 \\%& See also \cite{leveau04} \\
		20 & rock1													& 15 \\%& See also \cite{leveau04} \\
		21 & sax1													& 12 \\%& See also \cite{leveau04} \\
		22 & trumpet1												& 14 \\%& See also \cite{leveau04} \\
		23 & violin2												& 15 \\%& See also \cite{leveau04}
	\end{tabular}
	\label{appendix.table_onset_files}
\end{table}
\end{landscape}

\begin{figure}
	\centering
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n1.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n2.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n3.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n4.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n5.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n6.eps}
	\caption[Spectrograms of testfiles for onset detection (Testfiles 1 to 6).]{Spectrograms of testfiles for onset detection. \textbf{Left to right, top to bottom:} Testfiles 1 to 6, compare Table \ref{appendix.table_onset_files}.}
	\label{appendix.figure_onset_files1}
\end{figure}
\begin{figure}[htb]
	\centering
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n7.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n8.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n9.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n10.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n11.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n12.eps}
	\caption[Spectrograms of testfiles for onset detection (Testfiles 7 to 12).]{Spectrograms of testfiles for onset detection. \textbf{Left to right, top to bottom:} Testfiles 7 to 12, compare Table \ref{appendix.table_onset_files}.}
	\label{appendix.figure_onset_files2}
\end{figure}
\begin{figure}[htb]
	\centering
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n13.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n14.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n15.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n16.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n17.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n18.eps}
	\caption[Spectrograms of testfiles for onset detection (Testfiles 13 to 18).]{Spectrograms of testfiles for onset detection. \textbf{Left to right, top to bottom:} Testfiles 13 to 18, compare Table \ref{appendix.table_onset_files}.}
	\label{appendix.figure_onset_files3}
\end{figure}
\begin{figure}[htb]
	\centering
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n19.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n20.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n21.eps}
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n22.eps} \\
	\includegraphics[scale=0.42]{figures/experiments/onset/original/n23.eps}
	\caption[Spectrograms of testfiles for onset detection (Testfiles 19 to 23).]{Spectrograms of testfiles for onset detection. \textbf{Left to right, top to bottom:} Testfiles 19 to 23, compare Table \ref{appendix.table_onset_files}.}
	\label{appendix.figure_onset_files4}
\end{figure}

\end{appendix}